Author's response to reviews

Title: Study protocol: the Sleeping Sound with Attention-Deficit/Hyperactivity Disorder Project

Authors:

Emma Sciberras (emma.sciberras@mcri.edu.au)
Daryl Efron (daryl.efron@rch.org.au)
Bibi Gerner (bibi.gerner@mcri.edu.au)
Margot Davey (margot.davey@southernhealth.org.au)
Fiona Mensah (fiona.mensah@mcri.edu.au)
Frank Oberklaid (Frank.Oberklaid@rch.org.au)
Harriet Hiscock (harriet.hiscock@rch.org.au)

Version: 5 Date: 21 December 2010

Author's response to reviews: see over
Dear Editor,

Thank you for sending through the proofs for our manuscript entitled “Study protocol: the Sleeping Sound with Attention-Deficit/Hyperactivity Disorder Project”.

We have made some minor typographical changes to the manuscript:

- Page 1 – correcting Harriet Hiscock’s email address
- Page 2 – spacing
- Page 3 – spacing
- Page 6 – changed ‘Paediatric’ to Pediatric.
- Page 13 – spacing
- Page 16 – changed ‘7’ to seven.
- Page 17 – spacing

We are happy with this final version of the manuscript.

Thank you.

Yours sincerely,

Emma Sciberras, Daryl Efron, Bibi Gerner, Margot Davey, Fiona Mensah, Frank Oberklaid, Harriet Hiscock